Author’s response to reviews

Title: Cost-effectiveness of ward-based pharmacy care in surgical patients: protocol of the SUREPILL (Surgery & Pharmacy In Liaison) study

Authors:

Monica de Boer (m.deboer@amc.nl)
Maya A. Ramrattan (m.a.ramrattan@amc.nl)
Jordy J.S. Kiewiet (j.j.kiewiet@amc.nl)
Eveline B. Boeker (e.b.boeker@amc.nl)
Kim B. Gombert-Handoko (k.b.gombert-handoko@olvg.nl)
Nicolette A.E.M. van Lent-Evers (nv lent@diakhuis.nl)
Paul F. Kuks (p.f.kuks@amc.nl)
Marcel G.W. Dijkgraaf (m.g.dijkgraaf@amc.nl)
Marja A. Boermeester (m.a.boermeester@amc.nl)
Loraine Lie-A-Huen (l.lieahuen@amc.nl)

Version: 3 Date: 2 March 2011

Author’s response to reviews: see over
Date : 2 maart 2011
Subject : manuscript revision

Dear Dr. Patel,

Please find enclosed our revised manuscript entitled: ‘Cost-effectiveness of ward-based pharmacy care in surgical patients: protocol of the SUREPILL (Surgery & Pharmacy in Liaison) study’.

We made the following changes in our manuscript:

- Page 1, insertion of word ‘Acknowledgements’ and deletion of the word ‘Collaborators’ in the sentence: As listed under Collaborators.
- Page 2, second paragraph: insertion of word ‘Design’ in paragraph title.
- Page 7, title of section: insertion of word ‘Design’.
- Page 17, paragraph collaborators placed in the Acknowledgements paragraph.
- Page 24, table: removed vertical lines from table.

Thank you for accepting this manuscript for publication in BMC Health Services Research.

Yours sincerely,

Monica de Boer, PharmD on behalf of the Principal Investigators,
Marja Boermeester, GI surgeon and clinical epidemiologist
Loraine Lie-a-Huen, professor of clinical pharmacy

Academic Medical Centre
Department of Clinical Pharmacy, M01-223
PO Box 22660, 1100 DD Amsterdam
The Netherlands